Math 597/609: Cryptography 1

Prof. Burt Rosenberg Ungar 523, x–2575 Office hours: TBA.

MTH 597/609–U (Fall, 1995) Time: Tu., Th. 6:25–7:40 P.M. Room: CC 411

ffl Goals: We want to understand Cryptography and its applications well enough so that we can apply it creatively in new situations.

ffl Textbook: Cryptography: Theory and Practice, Douglas R. Stinson.

ffl Assignments: Assignments will be arranged to satisfy two audiences: “thinkers” and “doers”.

The doers will program, and the thinkers will prove theorems. Which is not to say that programs go without thinking, and proving goes without doing, only to provide each category with a reasonably complimentary labeling.

ffl Grades and Tests: A term project will count for half of the grade. The other half will be the assignments during the term.

ffl Other Interesting Books:

– The Codebreakers, D. Kahn, Macmillan.

– Applied Cryptography, Protocols, Algorithms and Source Code in C, B. Schneier, John Wiley and Sons.

– Modern Cryptology – A Tutorial, Giles Brassard, Springer-Verlag.

– Elementary Cryptanalysis: A Study of Ciphers and Their Solution by Helen Fouché Gaines, Dover.